Author's response to reviews

Title: Screening for latent tuberculosis infection among undocumented migrants in Swiss healthcare centres; a descriptive exploratory study

Authors:

Patrick Bodenmann (patrick.bodenmann@hospvd.ch)
Paul Vaucher (paul.vaucher@hospvd.ch)
Hans Wolff (hans.wolff@hcuge.ch)
Bernard Favrat (bernard.favrat@hospvd.ch)
Fanny de Tribolet (pointdeau@bluewin.ch)
Eric Masserey (eric.masserey@vd.ch)
Jean-Pierre Zellweger (jean-pierre.zellweger@hospvd.ch)

Version: 3 Date: 13 March 2009

Author's response to reviews:

BMC Infectious Disease
MS: 1212618762241204

Lausanne, March 13th 2009

Dear Editor,

We have proof read the document and modified the format of the manuscript as suggested.

Reference were corrected, lines were suppressed from tables, figures were cropped, title was reformatted with sentence case, keywords were suppressed and abbreviations were given without returning to the line.

Regards,

Paul Vaucher,
Patrick Bodemann,
Jean-Pierre Zellweger